Author's response to reviews

Title: Isolated loss of inferior pubic ramus: A case report.

Authors:

Aly Saber (alysaber@hotmail.com)

Version: 2 Date: 19 September 2007

Author's response to reviews:

Dear/BioMed Central

Thank YOU for good facilities and I'll try to correct the points described in YOUR e-mail.

I met this only case through out my realively long medical experience from 1980 to 2007. So, I want to share the interest of other experts.

This case report was just presented at the 25th Annual Summer Meeting of the Egyptian Society of Surgeons (ESS) in Collaboration with the 17th Annual Congress of the Pan Arab Association of Surgeons (PAS) which held in the period of September 5th 7th, 2007 at Helnan Palestine Hotel, Alex.Egypt.